Author’s response to reviews

Title: Post Traumatic Stress Disorder and Coping Strategies among Adult Survivors of Earthquake, Nepal

Authors:

Ishwari Adhikari Baral (adhikari.ishwari@gmail.com)

Bhagawati K.C (bhagwatikc2016@gmail.com)

Version: 3 Date: 22 Mar 2019

Author’s response to reviews:

Gerard Clarke (Reviewer 1)

Thank you so much for your review and comments. For English, I have made best possible final correction.

Irene Bighelli (Reviewer 2)

Thank you so much for your review and comment. Yes, the threshold of PTSD checklist (pag 7, line 19-21 was changed because in the first manuscript, it was a mistake due to typing error, i.e. I wrote 38 for 33. Furthermore, to make sure, I went through the original tool and also checked my datasheet, and the analysis was done according the instruction in the original tool, i.e. cut off 33 was taken to identify the survivors with PTSD and no PTSD. Sorry for the inconvenience.